%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Template for writing a paper included in ``Genome Informatics 1999'' %
% using LaTeX2e.                                                       %
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
\documentclass[11pt]{article}
%\usepackage{graphics}
\usepackage{graphicx}
%%%%%%%%  General Remarks  %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% 1. 11pt fonts should be used.
% 2. When special style files are used (e.g., epsf.sty), 
%    please send the style files together.
% 3. Even if color images are used, only black/white images 
%    will be used for publication.
% 4. If you want to include figures in your abstract, please 
%    use graphics.sty or graphicx.sty. You can take these macros
%    from our site in the same directory.(graphics.tar.gz)
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

%%%%%%%%  Do not change the following parameters  %%%%%%%%%%%
\setlength{\topmargin}{-1cm}
\setlength{\oddsidemargin}{-0.5cm}
\setlength{\textwidth}{17cm}
\setlength{\textheight}{24cm}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\pagestyle{empty}

%%
%% Title...
%%
\title{\bf A Sample LaTeX2e Format for\\
\bf the Tenth Workshop on Genome Informatics}


%% We do not use \author to display authors...
\author{}
\date{}

\begin{document}
\begin{thebibliography}{99}
%
% References should be arranged in alphabetical order of authors.
% Provide the titles of articles.
%
\bibitem{ble91}
Bowie, J. U., L\"{u}thy, R. and Eisenberg, D.,
A method to identify protein sequences that fold into a known
three-dimensional structure,
{\em Science}, 253:164--170, 1991.

\bibitem{ks83}
Kabsch, W. and Sander, C.,
Dictionary of secondary structure: pattern recognition of hydrogen-bonded
and geometric features,
{\em Biopolymers}, 22:2577--2637, 1983.

\bibitem{ki98}
Kobayashi, H. and Imai, H.,
Improvement of the A$^*$ algorithm for multiple sequence alignment,
{\em Genome Informatics 1998}, Universal Academy Press, 120--130, 1998.

\bibitem{nd70}
Needleman, S. B. and Wunsch, C. D.,
A general method applicable to the search for similarities in
the amino acid sequences of two proteins,
{\em J. Mol. Biol.}, 48:443--453, 1970.

\bibitem{si97}
Shibuya, T. and Imai, H., 
Enumerating suboptimal alignments of multiple biological sequences efficiently,
{\em Proc. Pacific Symposium on Biocomputing 1997}, World Scientifc, 409--420, 1997.

\bibitem{sw81}
Smith, T. F. and Waterman, M. S.,
Identification of common molecular subsequences,
{\em J. Mol. Biol.}, 147:195--197, 1981.

\bibitem{wgwz92}
Watson, J. D., Gilman, M., Witkowski, J. and Zoller, M.,
{\em Recombinant DNA} (Second Edition), W. H. Freeman and Company,
New York, 1992.
\end{thebibliography}

\maketitle
\thispagestyle{empty}
\vspace{-1.8cm}

%%
%% Authors...
%%
\begin{center}
\begin{tabular}[t]{c@{\extracolsep{2em}}c@{\extracolsep{2em}}c}
\bf        Ayumi Shinohara \footnotemark[1] &
\bf        Hideo Matstuda  \footnotemark[2] &
\bf        Satoru Miyano   \footnotemark[3] \\
        \small\tt ayumi@i.kyushu-u.ac.jp&
        \small\tt matsuda@ics.es.osaka-u.ac.jp&
        \small\tt miyano@ims.u-tokyo.ac.jp\\
\end{tabular}

\smallskip

\begin{small}
\begin{tabular}{rl}
\footnotemark[1]& \parbox[t]{13cm}{
                  Department of Informatics, Kyushu University, 
                  6-10-1 Hakozaki, Higashi-ku, Fukuoka 812-8581, Japan.} \\
\footnotemark[2]& \parbox[t]{13cm}{
                  Department of Informatics and Mathematical Science, 
                  Graduate School of Engineering Science, Osaka University, 
                  1-3 Machikaneyama, Toyonaka, Osaka 560-8531, Japan.} \\
\footnotemark[3]& \parbox[t]{13cm}{
                  Human Genome Center, Institute of Medical Science, 
                  University of Tokyo,
                  4-6-1 Shirokanedai, Minato-ku, Tokyo 108-8639, Japan.}
\end{tabular}
\end{small}
\end{center}

%%
%% End of Authors.
%%

\begin{abstract}
\em 
This is the abstract of a sample paper for the 8th Workshop on Genome
Informatics.  The workshop is intended to provide a forum for
researchers working in interdisciplinary areas between biological
sciences and computer science.
\end{abstract}

\section{Introduction}

Here is the introduction text.

The Workshop on Genome Informatics is organized by the Genome
Science Research Project funded by a Grant-in-Aid for Scientific
Research on Priority Areas from the Ministry of Education, Science,
Sports and Culture of Japan.

\section{Sample Section}

Of course you can create other sections if you want.


\subsection{Figures}
When you use a figure (Fig.~\ref{figure:sample} with a special style file,
please provide it.

\begin{figure}
\begin{center}
\includegraphics{sample.eps}
\end{center}
\caption{This is a sample figure.}
\label{figure:sample}
\end{figure}

Here are examples of citation \cite{ble91,ki98,ks83,nd70,si97,sw81,wgwz92}.
References should be arranged in alphabetical order of authors and
should be cited in the text.

\section*{Acknowledgments}
The section for acknolwedgements should not be numbered.


\end{document}




